

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Black, Sandra E.; Devereux, Paul J.; Lundborg, Petter; Majlesi, Kaveh

# Working Paper Learning to take risks? The effect of education on risktaking in financial markets

UCD Centre for Economic Research Working Paper Series, No. WP15/09

**Provided in Cooperation with:** UCD School of Economics, University College Dublin (UCD)

*Suggested Citation:* Black, Sandra E.; Devereux, Paul J.; Lundborg, Petter; Majlesi, Kaveh (2015) : Learning to take risks? The effect of education on risk-taking in financial markets, UCD Centre for Economic Research Working Paper Series, No. WP15/09, University College Dublin, UCD School of Economics, Dublin

This Version is available at: https://hdl.handle.net/10419/109726

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU

#### UCD CENTRE FOR ECONOMIC RESEARCH

WORKING PAPER SERIES

2015

#### Learning to Take Risks? The Effect of Education on Risk-Taking in Financial Markets

Sandra E Black, University of Texas Paul J Devereux, University College Dublin Petter Lundborg and Kaveh Majlesi, Lund University

WP15/09

April 2015

UCD SCHOOL OF ECONOMICS UNIVERSITY COLLEGE DUBLIN BELFIELD DUBLIN 4

### Learning to Take Risks? The Effect of Education on Risk-Taking in Financial Markets<sup>\*</sup>

Sandra E. Black Department of Economics University of Texas, Austin NHH, IZA and NBER sblack@austin.utexas.edu

Petter Lundborg Department of Economics Lund University IZA and CED Lund petter.lundborg@nek.lu.se Paul J. Devereux School of Economics and Geary Institute University College Dublin CEPR and IZA devereux@ucd.ie

Kaveh Majlesi Department of Economics Lund University Knut Wicksell Centre for Financial Studies kaveh.majlesi@nek.lu.se

March 2015

#### Abstract

We investigate whether acquiring more education when young has long-term effects on risktaking behavior in financial markets and whether the effects spill over to spouses and children. There is substantial evidence that more educated people are more likely to invest in the stock market. However, little is known about whether this is a causal effect of education or whether it arises from the correlation of education with unobserved characteristics. Using exogenous variation in education arising from a Swedish compulsory schooling reform in the 1950s and 1960s, and the wealth holdings of the population of Sweden in 2000, we estimate the effect of education on stock market participation and risky asset holdings. We find that an extra year of education increases stock market participation by about 2% for men but there is no evidence of any positive effect for women. More education also leads men to hold a greater proportion of their financial assets in stocks and other risky financial assets. We find no evidence of spillover effects from male schooling to the financial decisions of spouses or children.

<sup>\*</sup> The data used in this paper comes from the Swedish Interdisciplinary Panel (SIP) administered at the Centre for Economic Demography, Lund University, Sweden. We thank Helena Holmlund for generously sharing the reform coding with us.

There is a strong correlation between educational attainment and participation in financial markets—according to the 2010 Survey of Consumer Finances in the United States, 37.1% of households headed by college graduates participated in the stock market, while this fraction was only 5.6% among households headed by high school dropouts. Even after controlling for wealth and/or income, this correlation remains. But is this a problem? That depends on what is driving this observed difference. To the extent this difference reflects differences in preferences or other characteristics that are also correlated with education, we may see no reason for concern. However, if this is the result of a lack of information or financial acumen, it may suggest a role for public policy. Financial decision-making often involves an understanding of complicated issues that may limit the ability of less-educated adults to make smart investment decisions.<sup>1</sup> But could more education ameliorate this problem? More generally, what is the role of education in financial decision-making? Does more education actually change people's investment behavior? And, if yes, how?

A key difficulty with identifying the causal effect of education on investment behavior is that education is likely correlated with many unobserved individual characteristics that may be related to investment behavior, such as risk tolerance, IQ, and family background; this suggests that observed correlations may reflect these unobserved differences and not a causal relationship. To address this issue, we take advantage of an education reform in Sweden in the 1950s and 1960s that increased compulsory schooling from 7 to 9 years in different municipalities at different times. The change in schooling induced by this reform is thus uncorrelated with other individual characteristics related to wealth and asset allocation. Using administrative data that includes information on the wealth portfolios of the population of Sweden in 2000, we can

<sup>&</sup>lt;sup>1</sup> Christelis et al. (2010) and Grinblatt et al. (2011) show that portfolio choice is related to cognitive abilities.

identify the causal effect of the increase in education induced by the legislative change on the wealth portfolios of those born in the period around the law change. By further linking the data to the *children* of these men and women, we are also able to study whether any positive effects of schooling spill over to the next generation.

There are several channels through which education might affect the decision to participate in the equity market and the share of assets invested in equities. One possibility is that more education reduces risk aversion, which is considered a major factor in willingness to bear financial risk.<sup>2</sup> Another possible factor is through income; obtaining more education typically leads to higher earnings and greater wealth that then enables greater investments in risky financial assets. Finally, the cost of gathering and processing information about the risks and returns in the market might be lower for more educated individuals.<sup>3</sup> To the extent that education may increase financial acumen, this suggests that the large literature estimating the return to education in terms of earnings may actually underestimate the financial benefits from increasing educational attainment.

In addition to broadening our knowledge about the financial returns to education, this work contributes to our knowledge of the underlying determinants of equity market participation. Evidence suggests that there is substantial welfare loss from the decision not to hold stocks in one's portfolio; historically, stock market investment has had a high return compared to safer financial assets such as bonds and money market funds. As a result, it is important to understand

 $<sup>^{2}</sup>$  Standard models imply that, in a frictionless market, differences in risk preferences and wealth (if risk taking is a function of wealth) are the main sources of cross-sectional variation in the share in equities, (e.g., Samuelson, 1969; Merton, 1969).

<sup>&</sup>lt;sup>3</sup> See Bertaut (1998); Guiso et al. (2003); Hong et al. (2004); Bogan (2008); and Christiansen, Joensen, and Rangvid (2008). In markets with frictions, transaction and information costs may explain why many individuals do not invest in stocks or other financial assets (Haliassos and Bertaut, 1995; Vissing-Jorgensen, 2002).

why individuals may under-invest.<sup>4</sup>

We find a causal effect of education on men's investment decisions. Among men, more education increases the likelihood of stock market participation, the likelihood that he holds risky financial assets in general, and the proportion of his financial assets that he invests in stocks and other risky assets. The evidence also suggests that greater financial wealth is a possible mechanism underlying this relationship. However, we find no evidence that these effects persist across generations or that these effects spillover to spouses. There is no evidence of an effect of education on the financial decisions of women. These findings are robust to a variety of specification tests. To the extent that there may be substantial welfare loss when individuals choose not to hold stocks, these results suggest there may be a role for policy to influence financial risk-taking.<sup>5</sup>

The paper unfolds as follows. Section 2 discusses the related literature, while Section 3 describes the relevant institutional background. Section 4 describes the compulsory schooling reform and our data. Section 5 outlines our empirical strategy and Section 6 presents our results. Section 7 then concludes.

#### 2. Related Literature

Although it is not the primary focus of the research, several previous studies have documented a strong relationship between the level of education and equity market participation in both Sweden and the United States (Campbell, 2006; Calvet, Campbell, and Sodini 2007,

<sup>&</sup>lt;sup>4</sup> In addition, understanding the determinants of stock market investment also provides insight into the determinants of the distribution of wealth (Guvenen 2006), ownership of individual retirement accounts (Bernheim and Garrett 2003), and wealth effects on consumption (Dynan and Maki 2001).

<sup>&</sup>lt;sup>5</sup> Cocco et al. (2005) show that in calibrated life-cycle models the welfare loss from no stockholding is between 1.5 and 2 percent of consumption.

2009; Barnea et al. 2010 among others).<sup>6</sup> This positive relationship remains even in specifications with multiple controls.<sup>7</sup> However, despite the presence of a robust correlation between education and investment in risky financial assets, there is only limited work identifying the causal effect of education on equity holding.

There is also a small literature--primarily focused on distinguishing genetic versus environmental determinants of investment behavior--that uses differences in investment behavior among twins. These studies sometimes include education as a control variable in specifications with twin fixed effects, where the education estimates are based on differences in education within twin pairs. Using Swedish data, Barnea et al. (2010) show that the association between an advanced degree (college or graduate) and stock market participation drops (and in the case of a college degree becomes statistically insignificant) when they control for twin fixed effects, suggesting that the effect of education on participation in the stock market is significantly lower after controlling for genetic factors. Also using a panel of Swedish twins, Calvet and Sodini (2014) find that education is not significantly correlated with risky market participation and the risky share of financial assets once they control for yearly twin fixed effects. Unfortunately, while informative, twin studies are quite limited in their ability to determine the causal impact of any particular characteristic and are better suited for isolating the genetic versus environmental components.

<sup>&</sup>lt;sup>6</sup> In a related literature, financial literacy has also been shown to be correlated with stock market participation and portfolio diversification. Van Rooij, Lusardi, and Alessie (2011) find that those with low financial literacy are much less likely to invest in stocks and Guiso and Jappelli (2008) show that measures of financial literacy are strongly correlated with the degree of portfolio diversification. Hastings et al. (2013) provide a survey of the literature on the relationship between financial literacy and financial outcomes.

<sup>&</sup>lt;sup>7</sup> For example, using the U.S. Survey of Consumer Finances and controlling for wealth, income, age, race, and risk tolerance, Campbell (2006) reports a large and significant role for education in predicting whether a household owns public equity. The same study warns that education variables in demographic regressions could be endogenous and overstate the effects of exogenous increases in education on investment behavior.

Cooper and Zhu (2014) adopt yet another approach and use U.S. data to estimate a structural life-cycle model of the relationship between education and risky market participation. They conclude that education affects household finance mainly through increasing income and that higher educational attainment is associated with a lower stock market entry cost and a larger discount factor. However, this approach relies on functional form for identification as opposed to exogenous variation in education induced by a policy change.

Most closely related to our own work is that by Cole, Paulson, and Shastry (2014) who use variation in state compulsory schooling laws between 1914 and 1978 in the US to examine the effect of education on a variety of wealth and credit measures. Using data from the Census as well as the FRBNY Consumer Credit Panel/Equifax dataset, they provide compelling evidence that education affects investment income and reduces the probability that an individual declares bankruptcy. Using the Survey of Income and Program Participation (SIPP), they find that an additional year of education increases the probability of owning equities by 4 percentage points (statistically significant at the 10% level). Unfortunately, for outcomes on stock ownership, they are limited by relatively small sample sizes; in addition, recent research has suggested there are a number of limitations to using compulsory schooling laws in the U.S. for identification that are not relevant to the Swedish case. Our paper advances our understanding of the effect of education on stock ownership by using an arguably cleaner source of variation combined with population data from Sweden; in these data, we can observe total wealth, stock ownership, as well as information about the risk distribution of the assets.<sup>8</sup>

<sup>&</sup>lt;sup>8</sup> Stephens and Yang (2014) demonstrate that IV estimates using U.S. compulsory schooling laws often change sign and significance with the addition of region by year controls and so are not robust across reasonable specifications. In addition, Cole et al. cluster standard errors by state-year rather than by state and so do not allow for serial correlation at the state-level. Bertrand et al. (2004) show that this can lead to greatly underestimated standard errors and Black, Devereux, and Salvanes (2008) demonstrate this for U.S. compulsory schooling law analysis.

#### 3. Institutional Background

It is important to understand the institutional context for our study, particularly the role of pension programs as a component of savings. While the social welfare and pension systems are quite generous in Sweden, a large proportion of financial wealth is held outside of pension funds. Non-retirement wealth accounts for almost 84 percent of aggregate household financial wealth (Calvet et al 2007), and it is this form of wealth that is the focus of our study.

Relative to countries such as the U.S., Sweden's pension system would be considered quite generous. Sweden has a mix of public and private pension schemes, and individuals are allocated to different pension systems depending on the public or private sector affiliation and year of birth of the individual. The longer one works, the higher pension one receives. The retirement age is flexible and individuals can claim retirement benefits beginning at age 61.

In 2000, when we measure asset allocation, the public pension system almost entirely consisted of a national pension plan financed on a pay-as-you-go basis.<sup>9</sup> In addition, most people receive an occupational pension from their employer. According to the Swedish Pensions Agency, about 90% of employees receive some pension benefits from their employer as a condition of employment. On average, around 4.5% of the employee's salary is put into employer provided schemes (Thörnqvist and Vardardottir, 2014). Swedish residents also have tax incentives to invest in private pension savings that are only accessible after retirement. However, as mentioned earlier, individuals still hold a substantial fraction of their wealth in non-retirement

<sup>&</sup>lt;sup>9</sup> An individual account system known as the Premium Pension System (PPS) was introduced in 1999 but, because these funds were so new, investment in the PPS funds was very low when we are measuring asset allocation. Therefore, the existence of these funds is unlikely to have had any important effect on asset allocation.

wealth. In addition, stock market participation rates are higher in Sweden than in many other countries such as the United States (Guiso, Haliassos, and Jappelli, 2001).<sup>10</sup>

Because we examine both male and female investment behavior separately, it is important to understand whether there are incentives to transfer wealth holdings from one spouse to another. There do not appear to be any such incentives. In the event of a divorce, in the absence of a prenuptial agreement, all assets are split equally among spouses. Until 2006, there was a 1.5% tax on wealth above 1.5 million SEK for single tax filers and 3 million for a married couple filing jointly. The value of jointly owned assets was split between the two tax filers. Thus, there were no incentives for husbands and wives to strategically allocate assets between themselves in order to reduce their wealth tax bill.

Finally, people who face greater labor income risk may be less likely to choose risky financial portfolios. Consequently, the unemployment insurance system could potentially affect individuals' risk-taking behavior in financial markets; a more generous system could create an incentive to take more risk with one's portfolio. In 2000, while the formal replacement rate was at 80 per cent of wages, the effective replacement rate taking earnings-ceilings into account was around 65 percent. (Carling et al, 2001).<sup>11</sup> Because of this, it is difficult to imagine that high risky market participation in Sweden compared to many other countries can be explained by the "generous" unemployment insurance system.

<sup>&</sup>lt;sup>10</sup> There is also a guaranteed pension for those who have had little or no income from work, and the size of this guaranteed pension is based on how long the person has lived in Sweden. In 2000, the maximum guaranteed pension, which applies to those who have lived in Sweden for at least 40 years, is 2394 SEK per month (\$254) before taxes for those who are married, and 2928 SEK per month (\$311) for a single person. A tax rate of 30 percent is then applied.

<sup>&</sup>lt;sup>11</sup> There is an earnings ceiling above which no additional benefits are paid. In 1996, it was estimated that 75 percent of employees had monthly earnings exceeding the ceiling (Bharadwaj et al. 2014).

### 4. Empirical Strategy

Because education is correlated with unobservable characteristics that are also likely related to portfolio allocation, we use variation in educational attainment induced by a change in the compulsory schooling law to identify the causal effect of education on risky market participation.

#### The Compulsory Schooling Reform

The Swedish comprehensive primary schooling reform was implemented across municipalities at various times during the 1950s and 60s. A parliamentary committee first initiated the reform in 1948, and its most notable feature was an increase in the mandatory years of schooling from seven to nine years.<sup>12</sup> To facilitate an evaluation of the reform, it was implemented gradually across municipalities in a manner meant to be representative of the country's population and geography.<sup>13</sup> In 1949, 14 municipalities introduced the reform, and additional municipalities were added year by year. (Marklund 1981). In 1962, the parliament mandated that all municipalities implement the reform by 1969.<sup>14</sup>

In addition to an increase in the years of compulsory schooling, the reform changed a number of features of the public school system. In the pre-reform school system, students were tracked at grade 6, based on their performance; after the reform, this early tracking was abolished and students were instead integrated until 9<sup>th</sup> grade. In practice, however, the change was less dramatic, since students in the new system were able to choose between different types of courses

<sup>&</sup>lt;sup>12</sup> In a few larger cities, mandatory schooling was eight years before the reform.

<sup>&</sup>lt;sup>13</sup> Both before and after the reform, Swedish children normally started school during the calendar year in which they turned seven, meaning that compulsory schooling after the reform usually lasted until the age of 16.

<sup>&</sup>lt;sup>14</sup> In general, the reform was implemented in all school districts within a municipality. The exceptions were the three largest cities, Stockholm, Göteborg, and Malmö, where the reform was implemented in different school districts in different years. We use information on the parish the person grew up in to correctly allocate the reform variable for these cities.

and could self-track.<sup>15</sup> The reform also changed the national curriculum. English became a compulsory subject in reform schools and was taught beginning in the fifth grade. However, in 1955, even non-reform schools were required to make this change to the curriculum. Beyond this, the reform did not lead to any other changes in the total number of hours taught or to the distribution of hours designated to different subjects.<sup>16</sup>

There is a substantial literature that uses changes in the compulsory school laws in Sweden to study a variety of outcomes, including income, health, and crime. Meghir and Palme (2005) showed that the reform increased educational attainment and led to higher labor incomes. Holmlund et al. (2011) used the reform as an instrument for parental schooling, and found evidence of a causal effect of parent's educational attainment on child's educational attainment, and Lundborg et al. (2014) used a similar strategy to establish a positive effect of maternal education on the health and skills of sons at age 18. Finally, Meghir, Palme, and Schnabel (2012) use the Swedish reform to show that extra education reduced crime rates both for the individuals affected and for their children.<sup>17</sup> Thus, there is much evidence that the compulsory schooling reform had substantial and meaningful effects on the cohorts affected by it.

Data

<sup>&</sup>lt;sup>15</sup> The tracking before grade nine was abolished in 1962 but in Math and foreign languages, students were still able to choose between harder or easier classes. In a detailed description of the schooling reform, Marklund (1987, p. 180) notes that "the reform school between 1955 and 1960 conformed to a streaming system that in terms of routes was not too much different from the old parallel school with one common school route and one junior secondary school route."

<sup>&</sup>lt;sup>16</sup> Because it expanded compulsory schooling, the reform led to an increase in demand for teachers. This induced some schools in the early years of the reform to hire teachers who were not formally qualified. Over time, as several teacher colleges were opened, the shortage began to ease in the mid-60s (Marklund 1981). Municipalities were compensated by the government for the additional financial burden of hiring teachers and expanding school facilities. For a thorough overview of the schooling reform, see Holmlund (2008) and Lundborg et al. (2014).

<sup>&</sup>lt;sup>17</sup> Hjalmarsson et al. (2014) used the same reform to study the effect of education on adult crime.

We begin with a comprehensive dataset based on merged administrative registers that contains information on all Swedish citizens born between 1930 and 1980; this dataset includes information on educational attainment, municipality of residence, basic demographic information, and detailed wealth data.

In order to assign reform exposure status to individuals, we need to know in which municipalities individuals grew up. To do that, we link data from the 1960 and 1965 censuses that include information on municipality of residence. For cohorts born between 1943 and 1949, we use information from the 1960 census and for those born between 1950 and 1955 we use information from the 1965 census. We drop individuals who were born prior to 1943.<sup>18</sup>

In order to determine which individuals were exposed to the reform, we make use of a reform algorithm constructed by Helena Holmlund. Using birth year and municipality and parish of residence when growing up, the algorithm assigns a binary reform exposure variable to each individual in these cohorts. The algorithm is able to assign reform exposure to 90 percent of individuals born 1943-1955 who have non-missing information on municipality and parish of residence.<sup>19</sup>

In order to assign years of schooling to our sample, we use data from the education register in 1990. The register contains information on highest educational degree completed,

<sup>&</sup>lt;sup>18</sup> The reason for restricting our sample to those born 1943 and onwards is that the 1960 census provides a less accurate measure of municipality of residence for those born prior to 1943. A person born in 1940, for instance, may have moved out of his parent's place in order to work or study by the time of the census. Holmlund (2008) documents that the fraction of individuals living with their mothers in 1960 drops substantially for the pre-1943 cohorts.

<sup>&</sup>lt;sup>19</sup> Only 0.19 percent of the population living in Sweden at the time of the 1960 and 1965 censuses lack information on municipality of residence. The reform indicator is subject to measurement error. First, the reform exposure algorithm assumes that the students were in the right grade for their age. Svensson (2008) showed that 88 percent of all children born in 1949 were in the right grade in 1961, reflecting both that some students repeated a class and that some students started school earlier. Second, it is not always possible to assign a sharp starting date of the reform. These measurement problems only concern the cohorts born right around the assumed starting date of the reform and do not affect the consistency of the instrumental variables estimator we use.

which we use to impute years of schooling.<sup>20</sup> Table 1 shows the distribution of schooling in our sample 2 years before and 2 years after the reform.<sup>21</sup>

For data on asset allocation, we predominantly rely on the Swedish Wealth Data (Förmögenhetsregistret) from the year 2000. These data were collected by the government's statistical agency, Statistics Sweden, for tax purposes between 1999 and 2007, when the wealth tax was abolished.<sup>22</sup> The data includes all financial assets held outside retirement accounts at the end of a tax year, December 31st, reported by a variety of different sources, including the Swedish Tax Agency, welfare agencies, and the private sector. Financial institutions provided information to the tax agency on their customers' security investments and dividends, interest paid or received, and deposits.<sup>23</sup> Since the information is based on statements from financial institutions, it is likely to have very little measurement error and, since the entire population is observed, selection bias is not a problem.

In this paper, we have data on the aggregate value of bank accounts, mutual funds, stocks, options, bonds, and capital endowment insurance as well as total financial assets and total assets

<sup>&</sup>lt;sup>20</sup> We follow Holmlund et al. (2011) and impute years of schooling in the following way: 7 for (old) primary school, 9 for (new) compulsory schooling, 9.5 for (old) post-primary school (realskola), 11 for short high school, 12 for long high school, 14 for short university, 15.5 for long university, and 19 for a PhD university education. Since the education register does not distinguish between junior-secondary school (realskola) of different lengths (9 or 10 years), it is coded as 9.5 years. For similar reasons, long university is coded as 15.5 years of schooling.

<sup>&</sup>lt;sup>21</sup> In the table, there is no category for 8 years of schooling as the education register does not distinguish between different lengths of pre-reform primary school. Some school districts in the big cities had already implemented 8 years of mandatory schooling at the time of the reform.

<sup>&</sup>lt;sup>22</sup> The wealth tax used to be paid on all the assets of the household, including real estate and financial securities, with the exception of private businesses and shares in small public businesses (Calvet et al. 2007) and was levied at a rate of 1.5 percent on net household wealth exceeding SEK 900,000 in year 2000. The Swedish krona traded at \$0.106 at the end of 2000, so this threshold corresponds to \$95,400.

<sup>&</sup>lt;sup>23</sup> Importantly, nontaxable securities and securities owned by investors below the wealth tax threshold were included in the reports (Calvet et al. 2007).

for the population of Sweden.<sup>24</sup> For the analysis, we use data on the value of stocks, the value of risky assets (stocks plus mutual funds with a risky component), and total financial wealth held by individuals.

For data on income, we use the Income Register which includes income beginning in 1968. Our measure of earnings includes earnings from employed labor as well as selfemployment.

Our final sample for analysis includes more than 1.3 million individuals born between 1943 and 1955 for whom we have complete information on schooling, municipality and parish of residence when growing up, and wealth in 2000. While 1.8 million Swedish citizens were born between 1943 and 1955, twenty percent of those did not participate in the 1960 census; however, this group consists almost entirely of immigrants who had not arrived in Sweden by 1960 and, consequently, were not exposed to the educational reform. Out of the remaining individuals, 95 percent survived until 2000. Only 2 percent lack information in the education register of 1990.

We analyze equity market participation using four outcome variables constructed from the 2000 Wealth Register.<sup>25</sup> The first is an indicator variable for whether the individual owns stocks directly – we refer to this as stock market participation or direct equity participation. The second variable is an indicator variable for participation through either direct stock holding or mutual funds with a stock component – we refer to this as risky market participation.<sup>26</sup> Our final two

<sup>&</sup>lt;sup>24</sup> We have incomplete information on small bank accounts as they did not need to be reported by banks to the Swedish Tax Agency unless there was more than 100 SEK (about \$10) in interest during the year.

<sup>&</sup>lt;sup>25</sup> The choice of year 2000 is somewhat arbitrary. As a result, we later show that our results are quite similar when we use the years 1999, 2001, and 2002. An advantage of using data from the earlier years of the Wealth Register is that all individuals in the sample are below the age of retirement. Fagereng, Gottlieb and Guiso (2013) show a rebalancing of the portfolio away from stocks as investors approach retirement, and stock market exit after retirement.

<sup>&</sup>lt;sup>26</sup> This includes holdings of mutual funds that only include stocks but also includes mutual funds that have a mixture of stocks and other financial instruments such as bonds. By definition, persons with zero financial wealth are

measures are the share of financial assets held in stocks, and the share of financial assets held in equities, both conditional on participation. This latter variable is defined as the proportion of financial assets that are either in stocks or in mutual funds that have a stock component.

Table 2 provides summary statistics for our data. On December 31 2000, 36 percent of individuals in our sample directly held stocks outside retirement accounts, while 63 percent held equities either through direct stock holding or mutual funds with a stock component. Although there is no difference in men and women's rate of participation in risky financial markets, men held more of their risky assets as stocks while women preferred mutual funds with an equity component. Conditional on participation, the average equity share for men and women was 0.37 and 0.31 and the average risky share was 0.54 and 0.67 respectively.

#### 5. Empirical specification

Our empirical specification is based on the two following equations. Our first stage equation is:

$$S_{icm} = \pi_0 + \pi_1 R_{cm} + \theta_m + \delta_c + \varepsilon_{icm}, \tag{1}$$

where  $S_{icm}$  denotes the number of years of schooling of individual *i*, belonging to cohort *c*, and growing up in municipality *m*. Reform exposure,  $R_{cm}$  is a dummy variable taking the value of one if the individual was exposed to the reform.  $\theta_m$  and  $\delta_c$  denote municipality and birth cohort fixed effects.

The main equation of interest is:

$$FIN_{icm} = \gamma_0 + \gamma_1 S_{icm} + \theta_m + \delta_c + \varepsilon_{icm}, \qquad (2)$$

considered to have no holdings of stocks or risky assets.

where  $FIN_{icm}$  denotes our outcomes of interest, such as stock market participation or the share of risky assets. Here,  $S_{icm}$  is instrumented with reform exposure, according to Equation (1). The parameter of interest is  $\gamma_1$  which captures the causal effect of schooling on financial outcomes. Standard errors are clustered at the municipality level to allow for heteroskedasticity and arbitrary serial correlation across cohorts within municipalities. In addition to the basic specification in (1), we also estimate specifications with controls for parental education, with municipality-specific trends, and with region by cohort dummies. As we will see later, the estimates are quite robust to the exact choice of specification.

As described earlier, the reform was not randomly implemented across municipalities. Our empirical approach is therefore based on the assumption that, conditional on birth cohort fixed effects and municipality fixed effects (and, in some specifications, parental education, municipality-specific trends and region by cohort dummies), exposure to the reform is as good as random.

While we cannot test this assumption directly, we can provide evidence that suggests it is valid. We first examine whether the timing of reform implementation is related to observable characteristics; these results are presented in Table A1. The first column shows the relationship between parental schooling and our binary indicator of reform exposure without controlling for municipality dummies. Here, both mothers and fathers' schooling is positively and significantly associated with reform exposure, suggesting that, in a given year, the reform was more likely to be implemented in municipalities where the parents were better educated on average. However, when we include municipality fixed effects in column (2), this wipes out these significant correlations; the estimates of parental schooling in column (2) are tiny and statistically insignificant. This continues to be the case in the specifications with region by birth cohort fixed effects and municipality-specific trends (columns 3 and 4), suggesting that our identifying assumption may be reasonable.

Finally, a concern may be that some parents may have responded to the reform by moving to municipalities that were early implementers in order to ensure their child would benefit from the reform. Such endogenous mobility has previously been investigated by Meghir and Palme (2005) and by Holmlund (2008); both studies find little reason for concern. Only between 3 and 4 percent moved from a municipality that had not yet implemented the reform to a one that had, and an equal share moved in the opposite direction. In addition, mobility was not found to be systematically related to observable characteristics that are associated with education, such as parent's education.

#### 6. Results

#### First Stage

We first examine the relationship between exposure to the reform and years of education. In Table 3, we present the regression results of the first-stage effects of reform exposure on education using four different specifications. Column (1) includes controls for birth cohort fixed effects and municipality fixed effects; we find the reform increases education by 0.27 years for men and 0.16 years for women. To control for possible omitted variables, we next include additional controls for parental schooling to our first-stage regressions. Column (2) in Table 3 adds controls for mother's schooling; this has no effect on the estimates of the effect of the reform. When we next add region by year of birth fixed effects to the base specification (Column 3), this leads to a slight increase in the coefficients to 0.3 for men and 0.18 for women. Finally, column (4) adds municipality-specific linear trends to the base specification; once again, the addition of these controls leads to a small increase in the first stage estimates to 0.32 for men

and 0.21 for women. For both men and women, the F-statistics suggest that the instrument is sufficiently strong in all specifications.

#### Effects of Education on Stock and Risky Asset Holding

Table 4 shows the effect of schooling on individual stock market and risky asset participation. Panel A presents the estimates for stock market participation of men and panel B presents the corresponding results for women. The first column shows the OLS relationship between schooling and stock market participation. Not surprisingly, this estimate is positive and highly significant for both men and women, with magnitudes of 0.034 and 0.029 respectively, implying that an extra year of schooling is associated with about a 3% increase in the probability of stock market participation. Columns (2) to (5) then present the IV estimates for a variety of specifications. All specifications include cohort and municipality dummies and columns (3) - (5)also include controls for mother's schooling, region by year of birth fixed effects, and municipality-specific trends, respectively. Among males, in the baseline specification (Column 2), the estimate suggests that an extra year of education increases stock market participation about 2% from a base of 42%. Importantly, our results are quite robust to specification choice with estimates for men of about 2% in all columns.<sup>27</sup> Among women, although the OLS estimates are quite similar to those of men, the IV estimates provide no evidence for a positive effect of education on stock market participation. Indeed the point estimates are negative, albeit statistically insignificant.

In panels C and D of Table 4, we study a broader measure of risky financial behavior -risky market participation. This variable is one for people who either own stocks directly or who

 $<sup>^{27}</sup>$  Stephens and Yang (2014) show that, in the US, estimates of compulsory schooling are generally not robust to adding interactions of census region and year of birth to the regression. In column (4) we show that our estimates are robust to adding region by year of birth fixed effects. Sweden has 21 counties that we use to construct these fixed effects.

own stocks indirectly through mutual funds. The IV evidence of an effect of education is now weaker for men, with point estimates of about 1% that are marginal in terms of statistical significance. This suggests that the effect of education may in fact be working through the decision to purchase individual stocks instead of holding mutual funds with a stock component. The estimates for women continue to be negative and statistically insignificant.

In addition to influencing the participation decisions, education could also affect the allocation decision between risky and less risky assets. In Table 5, we look beyond the extensive margin and study the effects of education on the log (share) of stock holdings and risky assets in financial wealth, conditional on participation. Panels A and B show the results for the share of financial wealth directly invested in stocks and Panels C and D have the analogous estimates for log (share) of financial wealth in risky assets. For men, there is a positive and statistically significant effect of education on the log (share) of wealth held in stocks. The estimates imply that, conditional on participation, one more year of education results in around a 10 percent increase in the share of stocks in financial wealth for men. Relative to the average share of 37 percent, this is equivalent of around 3.7 percentage point increase in the share of stocks for men.<sup>28</sup> However, as at the extensive margin, there is no evidence of any effect for women. Also, as was the case at the extensive margin, when we turn to the share of financial wealth in risky assets (stocks and stock-containing mutual funds), the effects for men are positive but smaller (and less statistically significant) than for stock holding. As before, there is no evidence of an effect for women.

 $<sup>^{28}</sup>$  As is common in the literature on the returns to education, the IV estimates here are larger than the OLS estimates despite priors that OLS is likely biased upwards due to omitted variable bias. This can be explained by a variety of factors, the most likely being that the IV represents the local average treatment effect—the effect of education for those who would have obtained 7 years of education but who, after the reform, obtain 9. The effects of education on this sample may be larger than the average effects for the population.

## Robustness Checks

We also conducted a number of robustness checks; these are presented in Appendix Tables 2 and 3. For parsimony, we only report estimates for our baseline specification with municipality and cohort fixed effects. One possible concern is that we are including too many years before and after the reform and should look more closely around the law change. Panel A of Table A2 shows our main results when we include only cohorts born within 5 years of the first reform cohort in that particular municipality to verify that we are making appropriate comparisons; the results are quite robust to this restriction. Because of uncertainty regarding the exact number of pre-reform years of schooling in the three cities of Stockholm, Göteborg, and Malmö, Panel B, presents estimates when we re-run our main specification dropping these cities. The estimates are similar but the standard errors fall somewhat. This probably arises because the first stage is weaker due to the generally higher levels of pre-reform education in these cities.<sup>29</sup>

Finally, Appendix Table 3 presents results when we consider different years of data. Our decision to use wealth data from 2000 was arbitrary, and Appendix Table 3 shows our results are robust to this choice by reporting estimates using data from 1999, 2000, 2001, and 2002 in columns (1) to (4) respectively. The last column then shows estimates using averages of the dependent variable for each individual across the 4 years. These estimates are almost identical to the estimates using 2000 data; this is unsurprising as investment behavior tends to change slowly over time.

#### Intra-Household Spillovers

So far, we have considered the role of individual education on his/her own investment behavior. However, when one considers families, it may be that one spouse influences the

 $<sup>^{29}</sup>$  Excluding these three cities increases the first stage coefficient in the baseline specification from 0.27 to 0.34 for males and from 0.16 to 0.21 for females.

other's investment behavior.<sup>30</sup> To test this empirically, we can estimate the effects of one spouse's education on the portfolio allocation of his/her spouse.<sup>31</sup>

In Table 6, we look for spillovers within households directly by studying how one spouse's education affects the asset holding of the other. We do this with and without a control for the education of the other spouse, where we instrument both own education and spousal education with reform exposure. We find no evidence of any spillover effects.<sup>32</sup>

### Intergenerational Spillovers

Finally, because of the richness of our data we are able to examine whether the effects of education on financial risk-taking are transmitted to the children. To do so, we estimate similar specifications, but now our outcome measures are the risk allocation of the *children* of the affected cohorts.

In order to link children to their parents, we make use of the Swedish multigenerational register. With our data, we are able to link children born 1980 or earlier to our main sample born 1943-1955. For our female sample, we observe 920,148 children distributed across 492,224 mothers. For fathers, we observe 713,886 children among 404,982 fathers.<sup>33</sup>

<sup>&</sup>lt;sup>30</sup> As we noted earlier, there is no tax incentive for spouses to reallocate assets from one to the other.

<sup>&</sup>lt;sup>31</sup> For married people, we can also examine the effect of their education level on the investment decisions of the household, not just their individual investments. When we do this, we continue to find evidence of an effect of education on individual investment behavior within two-adult households (even in this reduced sample) but find no evidence of an effect on total household asset allocation.

<sup>&</sup>lt;sup>32</sup> The reason for the smaller number of observations in these regressions is that the sample is restricted to cases where also the spouse has non-missing information on education, birth year, and reform exposure. In these analyses, we put no restriction on the birth year of the spouse.

<sup>&</sup>lt;sup>33</sup> We thus lack information on children for 23 percent of the women, which reflects that some women remain childless throughout their life and that some children were born after 1980. From external sources we know that about 12 percent of women born during our study period remained childless (SCB 2011). We lack information on children for 39 percent of our males, where the higher rate reflects the fact that a higher proportion of males in our

In order to study outcomes of the children at the oldest possible age, we focus on data from the wealth register as of 2006. This is the last year of data in the register; at this point, the youngest child is at least 26, which is well past the age at which most young Swedish adults have completed their education. Table A4 shows summary statistics for the sample of children, revealing that the average age of the children in 2006 was 33. Despite their relatively young age, almost half of the children hold some risky asset and about a fifth hold stocks directly.

Table 7 summarizes the results, using the previous IV specification that includes municipality and birth cohort fixed effects. Panel A shows the effects for men. In the first column, the estimate of the effect of father's schooling on children's stock market participation is 0.002 and insignificant. We get similarly tiny and insignificant estimates for children's risky market participation and for their log share of stocks or risky assets. The corresponding results for mothers, shown in Panel B of Table 7, are small and insignificant in all cases, suggesting that mothers' schooling does not causally influence children's financial holdings. This is in contrast to the OLS estimates that suggest significant and positive correlations between parental schooling and children's holdings.<sup>34</sup>

There are two potential concerns with the analysis above. First, parental schooling may be related to the probability of observing a child in our data, which could bias our results. This would be the case if schooling increases the age at first birth or the probability of remaining childless. We can test this directly; to do so, we estimate the effect of schooling on the probability of not observing any child our data. Doing so, we get positive but statistically insignificant effects of schooling for both males and females, suggesting no systematic

study cohorts remain childless and that, on average, males have their first child later than females (SCB 2011).

<sup>&</sup>lt;sup>34</sup> This finding appears inconsistent with recent U.S. evidence that used variation induced by compulsory schooling laws and data from the Panel Study of Income Dynamics that found a negative effect of parental education on the measured risk aversion of their children (Hryshko et al. 2011).

differences in the probability of being in our sample by parental education.

A second concern is that, among the parents where we observe at least one child, schooling might be negatively related to the age of the child in 2006. If younger children hold fewer assets, this might bias us against finding an effect of schooling. To test this, we estimate the effect of schooling on age of the child in 2006. When we do this, we again find small and insignificant effects for both males and females.<sup>35</sup>

#### Discussion

We have found that exogenous increases in education lead to greater stock market participation among men. This is an important finding, as it suggests that risk-taking behavior in financial markets is partly determined by educational policies. However, the question remains: what are the mechanisms underlying the effect of male education on investment behavior?

While we cannot definitively answer this question, we next explore some of the potential mechanisms through which education could operate. One channel, emphasized by Cooper and Zhu (2014), is that education increases earnings and, hence, leads to greater risky market participation. This could be because higher stable return to human capital can partially substitute for bond holding, or perhaps because the fixed costs of investment decrease with earnings.

For comparison to the existing literature, we first estimate the returns to education using the compulsory schooling instrument and our sample. In this case, the dependent variable is the log of average earnings between 1980 and 2000, including only those years with positive earnings. These results are presented in column (1) of Table A5. Using 2SLS, the return to education is about 3% for men but there is no evidence of a positive return for women.<sup>36</sup> It could

 $<sup>^{35}</sup>$  The estimates are -0.07 for males and -0.17 for females. In addition, we have run the analyses in Table 7 controlling for the age of the child in 2006. The conclusions were robust to this change.

<sup>&</sup>lt;sup>36</sup> The OLS estimates suggest a return of about 7% for both men and women. The low 2SLS earnings returns to

be that education affects earnings for men, and these increased earnings affect their risk-taking behavior. As a crude way of testing this channel, columns (2) to (5) of Table A5 include controls for log(earnings). As a result, there are falls in the size of the coefficients of education on risk taking but the changes are not large, suggesting that the earnings channel may not be very important.

Another possible mechanism is that the changes in portfolio allocation are due to differences in financial wealth.<sup>37</sup> While education could increase financial wealth through its effect on earnings, it could also affect it directly by changing saving or consumption behavior. If people become less risk averse as their wealth increases, we would expect to see increased wealth due to increases in education leading to changes in financial risk-taking.<sup>38</sup> As with labor earnings, we next include log(financial wealth) as a control in our regressions; we can then attribute the changes in our coefficients of interest as the fraction "explained" by differences in financial wealth.<sup>39</sup> We also report 2SLS estimates of the effect of education on log(financial wealth).

In order to consider financial wealth as a possible mechanism, one much first deal with the issue that many people (24%) are recorded as having zero financial wealth. This arises because, as mentioned earlier, small bank accounts are underrepresented in our data. Since almost

education are consistent with estimates from previous Swedish studies (Meghir and Palme 2005). Small earnings returns to education are common findings in European data and have recently been found for Britain (Devereux and Hart, 2010) and Germany (Pischke and von Wachter, 2008).

<sup>&</sup>lt;sup>37</sup> In a related literature, Behrman et al. (2012) show that greater financial literacy leads to higher wealth accumulation in Chile.

<sup>&</sup>lt;sup>38</sup> However, even with constant relative risk aversion (CRRA), more education could lead to greater risk tolerance and greater financial wealth through higher returns. In this scenario, greater financial wealth is a result of greater risk-taking rather than a mechanism that leads to greater risk taking.

<sup>&</sup>lt;sup>39</sup> It is important to note that identifying the role of financial wealth is murkier than identifying the role of income in the case of financial wealth, changing risk-taking behavior will itself affect financial wealth; income does not have this same endogeneity problem.

everybody has a bank account, in reality the people who we measure as having zero financial wealth probably in fact have some small amount of financial wealth.<sup>40</sup> In addition, the financial wealth distribution is extremely skewed, with a small number of individuals having very large holdings. For these reasons, we use the log of financial wealth in the analysis and we impute financial wealth as being 1 for cases with zero reported wealth.<sup>41</sup> To verify that our conclusions are not being driven by outliers, we also conduct robustness checks where we trim the top of the financial wealth distribution so as to reduce the impact of individuals with extremely high financial wealth.

Table 8 presents the results using our baseline specification with cohort and municipality fixed effects along with controls for financial wealth. As we can see, the addition of a control for financial wealth has a substantial effect on the IV estimates. For example the effect of education on the stock market participation of men falls from 0.018 (Table 4) to 0.012 with the addition of the control for log financial wealth. The last column of Table 8 directly estimates the effect of education on log financial wealth. The IV estimates are statistically insignificant for both men and women. However the point estimate for men (11%) is quite substantial. As noted earlier, one concern is that these estimates are strongly influenced by outliers with very large financial wealth. In Table A6, we show that omitting observations with the top 5% or 10% of financial wealth leads to a larger and more precisely estimated statistically significant 18% effect of a year of education on financial wealth. Overall, the evidence suggests that education increases financial

<sup>&</sup>lt;sup>40</sup> In surveys, the fraction of Swedes aged 15 and above that have a bank account has consistently been 99 percent (Riksbanken, 2014).

<sup>&</sup>lt;sup>41</sup> Using Inverse Hyperbolic Sine function instead, that is defined for zero financial wealth, generates identical results.

wealth substantially and that this is probably an important mechanism through which education increases stock market participation.<sup>42</sup>

Another possible mechanism is that effects of education on portfolio allocation are due to the effects of education on attitudes towards risk-taking. While information deficits and fixed costs are plausible reasons for why education may increase the likelihood of investing in stocks, risk aversion may be a more likely explanation for why education increases the proportion of financial wealth invested in stocks, conditional on participation.<sup>43</sup> Our findings for men that education increases risk-taking at both the extensive and intensive margins are consistent with an effect of education on risk-attitudes over and above any effects on information and fixed costs.<sup>44</sup>

#### 7. Conclusion

By using the increased educational attainment induced by the change in the compulsory schooling legislation in Sweden in combination with a rich dataset containing wealth information for the entire population of the country, we are able to estimate the causal relationship between education and risk-taking in the financial market. We find evidence of a positive effect of education on stock holding for men but not for women. Interestingly, we find no evidence of spillovers from husbands to wives; husband's education does not affect the stock market participation behavior of his wife and vice-versa. We also find no evidence that higher education of men leads to their children being more likely to hold a higher proportion of financial wealth in

<sup>&</sup>lt;sup>42</sup> Interestingly, adding log(earnings) as an control in addition to log(financial wealth) has no effect on the effect of education on risk-taking, suggesting that the effects of income probably work through their effects on financial wealth.

<sup>&</sup>lt;sup>43</sup> In a standard asset pricing model (assuming CRRA and Independently and Identically Distributed returns), the risk preference parameter for an individual is proportional to the share that the individual invests in equities.

<sup>&</sup>lt;sup>44</sup> However, Jung (2015) finds using UK survey data and compulsory schooling laws that self-reported risk aversion is increased by extra schooling.

risky assets. These results are robust to a variety of specifications and a number of robustness checks. The evidence also suggests that greater financial wealth is a possible mechanism underlying this relationship. Given existing research suggesting substantial welfare costs to choosing not to hold stocks (Cocco et al. 2005), these findings suggest a role for government policy; this is yet one more benefit of increasing educational attainment.

# References

Barnea, A, Cronqvist, H, and S. Siegel (2010). Nature or Nurture: What determines investor behavior? *Journal of Financial Economics*, 98: 583-604.

Behrman, J.R., O.S Mitchell, C.K. Soo, and C. Bravo (2012). How Financial Literacy Affects Household Wealth Accumulation. *American Economic Review*, 102(3): 300-304.

Bernheim, B. D., & Garrett, D. M. (2003). The effects of financial education in the workplace: evidence from a survey of households. *Journal of Public Economics*, *87*(7), 1487-1519.

Bertaut, C. C. (1998). Stockholding behavior of US households: evidence from the 1983–1989 survey of consumer finances. *Review of Economics and Statistics*, 80(2), 263-275.

Bertrand, Marianne, Esther Duflo and Sendhil Mullainathan, 2004, How much should we trust differences- in-differences estimates?, *Quarterly Journal of Economics* 119, 249-275

Bharadwaj, P., Lundborg, P., & Rooth, D. O. (2014). Health Endowments and Unemployment during Macroeconomic Crises. Manuscript

Black, S. E., Devereux, P. J., & Salvanes, K. G. (2008). Staying in the Classroom and out of the maternity ward? The effect of compulsory schooling laws on teenage births. *The Economic Journal*, *118*(530), 1025-1054.

Bogan, V. (2008). Stock market participation and the internet. *Journal of Financial and Quantitative Analysis*, 43(01), 191-211.

Calvet, L. E., Campbell, J. Y., & Sodini, P. (2007). Down or Out: Assessing the Welfare Costs of Household Investment Mistakes. *Journal of Political Economy*, *115*(5), 707-747.

Calvet, L. E., Campbell, J. Y. and Sodini, P. (2009). Fight Or Flight? Portfolio Rebalancing by Individual Investors. *The Quarterly Journal of Economics*, 124.

Calvet, L. E., & Sodini, P. (2014). Twin Picks: Disentangling the Determinants of Risk-Taking in Household Portfolios. *The Journal of Finance*, *69*(2), 867-906.

Campbell, J. Y. (2006). Household finance. The Journal of Finance, 61(4), 1553-1604.

Carling, K., Holmlund, B., & Vejsiu, A. (2001). Do benefit cuts boost job finding? Swedish evidence from the 1990s. *The Economic Journal*, *111*(474), 766-790.

Christelis, D., Jappelli, T., & Padula, M. (2010). Cognitive abilities and portfolio choice. *European Economic Review*, 54(1), 18-38.

Christiansen, C., Joensen, J. S., & Rangvid, J. (2008). Are economists more likely to hold stocks? *Review of Finance*, *12*(3), 465-496.

Cocco, J. F., Gomes, F. J., & Maenhout, P. J. (2005). Consumption and portfolio choice over the life cycle. *Review of financial Studies*, 18(2), 491-533.

Cole, S., Paulson, A., & Shastry, G. K. (2014). Smart Money? The Effect of Education on Financial Outcomes. *Review of Financial Studies*, 27(7), 2022-2051.

Cooper, R., & Zhu, G. (2014). Household Finance over the Life-Cycle: What does Education Contribute? (No. w20684). National Bureau of Economic Research.

Devereux, Paul J. and Robert A. Hart. (2010). "Forced to be Rich? Returns to Compulsory Schooling in Britain." *Economic Journal* 120:1345–1364.

Dynan, K. E., & Maki, D. M. (2001). Does stock market wealth matter for consumption?

Fagereng, A., Gottlieb, C., & Guiso, L. (2013). Asset market participation and portfolio choice over the life-cycle. Netspar Discussion Paper 26/13.

Grinblatt, M., Keloharju, M., & Linnainmaa, J. (2011). IQ and stock market participation. *The Journal of Finance*, *66*(6), 2121-2164.

Guiso, L., Haliassos, M., & Jappelli, T. (2001). Household portfolios: An international comparison. *Household Portfolios*.

Guiso, L., Haliassos, M., & Jappelli, T. (2003). Household stockholding in Europe: where do we stand and where do we go? *Economic Policy*, *18*(36), 123-170.

Guiso, L., & Jappelli, T. (2008). Financial literacy and portfolio diversification. EIEF WP 12/08.

Guvenen, F. (2006). Reconciling conflicting evidence on the elasticity of intertemporal substitution: A macroeconomic perspective. *Journal of Monetary Economics*, *53*(7), 1451-1472.

Hjalmarsson, R., Holmlund, H, Lindqvist, R. (2014) "The Effect of Education on Criminal Convictions and Incarceration: Causal Evidence from Micro-data" Forthcoming, *Economic Journal*.

Haliassos, M., & Bertaut, C. C. (1995). Why do so few hold stocks? *The Economic Journal*, 1110-1129.

Hastings, J. S., Madrian, B. C., & Skimmyhorn, W. L. (2013). Financial Literacy, Financial Education and Economic Outcomes. *Annual Review of Economics*, 5, 347.

Holmlund, H. (2008). *A Researcher's Guide to the Swedish Compulsory School Reform. CEE DP* 87. Centre for the Economics of Education. London School of Economics and Political Science, Houghton Street, London, WC2A 2AE, UK.

Holmlund, H., Lindahl, M., & Plug, E. (2011). The causal effect of parents' schooling on children's schooling: a comparison of estimation methods. *Journal of Economic Literature*, 49(3), 615-651.

Hong, H., Kubik, J. D., & Stein, J. C. (2004). Social interaction and stock-market participation. *The Journal of Finance*, 59(1), 137-163.

Hryshko, D., M. J. Luengo-Prado and B. E. Sorensen, "Childhood determinants of risk aversion: The long shadow of compulsory education," *Quantitative Economics* 2 (03 2011), 37–72.

Jung, S. "Does Education affect risk aversion? Evidence from the British Education Reform" Applied Economics, forthcoming 2015.

Lundborg, Petter, Anton Nilsson, and Dan-Olof Rooth. (2014). Parental Education and Offspring Outcomes: Evidence from the Swedish Compulsory School Reform. *American Economic Journal: Applied Economics*, 6(1), 253-78.

Marklund, S. (1981). Från reform till reform: Skolsverige 1950–1975, Del 2, Försöksverksamheten. Stockholm: Skolöverstyrelsen och UtbildningsFörlaget.

Marklund, S. (1987). Från reform till reform: Skolsverige 1950–1975, Del 5, Försöksverksamheten. Stockholm: Skolöverstyrelsen och UtbildningsFörlaget.

Meghir, C., & Palme, M. (2005). Educational reform, ability, and family background. *American Economic Review*, 414-424.

Meghir, C., Palme, M., & Schnabel, M. (2012). The effect of education policy on crime: an intergenerational perspective. *National Bureau of Economic Research, (No. w18145).* 

Merton, R. C. (1969). Lifetime portfolio selection under uncertainty: The continuous-time case. *The Review of Economics and Statistics*, 247-257.

Pischke, J.-S. and von Wachter, T. (2008). Zero Returns to Compulsory Schooling in Germany: Evidence and Interpretation. *Review of Economics and Statistics* 90(3), 592-598.

Riksbanken (2014). Har kontanter någon framtid som lagligt betalningsmedel? Ekonomiska kommentar. Nr 9 2014.

Samuelson, P. A. (1969). Lifetime portfolio selection by dynamic stochastic programming. *The Review of Economics and Statistics*, 239-246.

SCB (2011). Olika generationers barnafödande. Demografiska rapporter 2011:3. SOU (2005). Svårnavigerat? Premiepensionsparande på rätt kurs. Betänkande av Premiepensionsutredningen. Stockholm: Statens Offentliga Utredningar (SOU) 2005:87.

Stephens Jr, M., & Yang, D. Y. (2014). Compulsory Education and the Benefits of Schooling. *American Economic Review*, *104*(6), 1777-92.

Svensson, A. (2008). Har dagens tonåringar sämre studieförutsättningar? En studie av förskjutningar i intelligenstestresultat från 1960-talet och framåt. Pedagogisk Forskning i Sverige 13 (4): 258–77.

Thörnqvist, T., & Vardardottir, A. (2014). Bargaining over Risk: The Impact of Decision Power on Household Portfolios. *Manuscript*.

Van Rooij, M., Lusardi, A., & Alessie, R. (2011). Financial literacy and stock market participation. *Journal of Financial Economics*, 101(2), 449-472.

Vissing-Jorgensen, A. (2002). Towards an explanation of household portfolio choice heterogeneity: Nonfinancial income and participation cost structures. National Bureau of Economic Research (No. w8884).

Years of schooling	Before	After
7	15.8%	2.3%
9	8.4%	20.6%
9.5	3.3%	0.5%
11	32.2%	34.6%
12	12.2%	12.1%
14	3.7%	4.4%
15.5	23.7%	24.9%
19	0.6%	0.6%
Ν	183,568	151,456

Table 1. Distribution of years of schooling two years before and after the reform.

Notes: Before indicates education distribution of cohorts in the two years prior to the reform, while After indicates the distribution of those two years post reform. Note that because the reform occurred in different municipalities at different times, the actual year of the reform varies by municipality.

		Males			Females	
	Mean	Standard Deviation	Obs	Mean	Standard Deviation	Obs
Portfolio characteristics:						
Financial wealth*	266,225	1,210,262	662,057	186,596	753,577	642,110
Direct equity participation**	0.42	0.49	662,096	0.30	0.46	642,119
Direct equity share***	0.37	0.34	275,389	0.31	0.32	194,618
Risky market participation**	0.63	0.48	662,096	0.63	0.48	642,119
Risky share***	0.54	0.39	417,957	0.67	0.33	403,705
Financial characteristics:						
Real Estate wealth*	656,229	2,184,561	662,096	459,735	1,315,905	642,119
Total wealth*	1030242	34,400,000	662,096	674,205	7,461,090	642,119
Total liabilities*	306,904	1,430,661	662,096	183,937	996,811	642,119
Demographic characteristics:						
Age in year 2000	51.17	3.72	662,096	51.20	3.72	642,119
Education	11.34	2.88	662,096	11.66	2.79	642,119
Married	0.60	0.49	662,096	0.62	0.49	642,119
Born in Sweden	0.97	0.16	662,096	0.97	0.17	642,119

Table 2. Summary Statistics

Notes: \* All monetary values are reported in Swedish Krona on December 31, 2000. At the time, the exchange rate was 1 USD = 9.42 SEK.

\*\* A dummy, taking a value of 1 if the individual participates.

\*\*\* Conditional on participation

	(1)	(2)	(3)	(4)
-		Panel	A: Males	
Reform exposure	0.266	0.266	0.305	0.323
	(0.043)***	(0.039)***	(0.023)***	(0.024)***
Ν	662,096	662,096	662,096	662,096
F-stats	38.79	46.42	177.67	186.33
		Panel	B: Females	
Reform exposure	0.164	0.163	0.183	0.213
-	(0.031)***	(0.029)***	(0.018)***	(0.020)***
Ν	642,119	642,119	642,119	642,119
F-stats	27.18	30.40	106.76	114.63
Mother's schooling	NO	YES	NO	NO
Region by cohort FE	NO	NO	YES	NO
Mun. trends	NO	NO	NO	YES

Table 3. First-stage regressions. Males and Females.

Notes: Columns (1)-(4) show the effect of reform exposure on years of schooling from specifications including birth cohort and municipality fixed effects. In addition, columns (2)-(4) include: (2) controls for mothers' schooling and an indicator of missing information on mothers' schooling, (3) region-by-county fixed effects, and (4) municipality-specific linear trends. Standard errors are clustered at the municipality level. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

	(1)	(2)	(3)	(4)	(5)
	Р	anel A: Stock	r market parti	cipation, male	es.
Schooling	0.034	0.018	0.019	0.022	0.016
	(0.001)***	(0.008)**	(0.009)**	(0.009)**	(0.008)**
Ν	662,096	662,096	662,096	662,096	662,096
	Pa	nel B: Stock	market partic	ipation, femal	es
Schooling	0.029	-0.018	-0.019	-0.025	-0.012
	(0.000)***	(0.017)	(0.016)	(0.014)*	(0.011)
Ν	642,119	642,119	642,119	642,119	642,119
	Р	anel C: Risky	<sup>,</sup> market parti	cipation, male	25
Schooling	0.033	0.012	0.012	0.020	0.012
-	(0.001)***	(0.008)	(0.008)	(0.008)**	(0.007)*
Ν	662,096	662,096	662,096	662,096	662,096
	Pa	nel D: Risky	market partic	ipation, femal	les
Schooling	0.036	-0.021	-0.021	-0.014	-0.016
	(0.000)***	(0.018)	(0.017)	(0.013)	(0.011)
Ν	642,119	642,119	642,119	642,119	642,119
Mother's schooling	NO	NO	YES	NO	NO
Region by cohort FE	NO	NO	NO	YES	NO
Mun. trends	NO	NO	NO	NO	YES

Table 4. Effect of education on participation in stock markets and risky markets. Males and females.

Notes: (1) OLS regression on the relationship between schooling and stock market participation (panels A-B) or risky market participation (panels C-D). Columns (2)-(5) show instrumental variables estimates of the effect of schooling from specifications including birth cohort and municipality fixed effects. In addition, columns (3)-(5) include: (3) controls for mothers' schooling, (4) region-by-county fixed effects, and (5) municipality-specific linear trends. Standard errors are clustered at the municipality level. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

	(1)	(2)	(3)	(4)	(5)
		Panel A: (le	og)share of ste	ocks, males	
Schooling	0.025	0.103	0.104	0.092	0.085
-	(0.002)***	(0.049)**	(0.050)**	(0.048)*	(0.045)*
Ν	275,389	275,389	275,389	275,389	275,389
		Panel B: (lo	g)share of sto	cks, females	
Schooling	0.010	0.003	-0.012	0.012	-0.038
-	(0.002)***	(0.094)	(0.106)	(0.075)	(0.077)
Ν	194,618	194,618	194,618	194,618	194,618
		Panel C: (log)	share of risky	assets, males	5
Schooling	0.016	0.048	0.048	0.023	0.035
	(0.001)***	(0.027)*	(0.027)*	(0.025)	(0.024)
Ν	417,957	417,957	417,957	417,957	417,957
	Р	anel D: (log)	share of risky	assets, female	25
Schooling	0.012	0.023	0.023	0.003	0.013
	(0.001)***	(0.037)	(0.039)	(0.038)	(0.029)
Ν	403,705	403,705	403,705	403,705	403,705
Mother's schooling	NO	NO	YES	NO	NO
Region by cohort FE	NO	NO	NO	YES	NO
Mun. trends	NO	NO	NO	NO	YES

Table 5. Effect of education on (log)share of stocks and risky holdings out of total financial wealth. Males and females.

Notes: (1) OLS regression on the relationship between schooling and share of stocks (panels A-B) and share of risky assets (panels C-D) out of financial wealth. Columns (2)-(5) show instrumental variables estimates of the effect of schooling from specifications including birth cohort and municipality fixed effects. In addition, columns (3)-(5) include: (3) controls for mothers' schooling, (4) region-by-county fixed effects, and (5) municipality-specific linear trends. The estimates are conditional on holding any stock (panels A-B) or any risky asset (panels C-D). Standard errors are clustered at the municipality level. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

	()	1)	(2	2)	(3)		(4	4)
	Stock market	participation	Risky market	participation	(log)share	e of stocks	(log)share o	f risky assets
				Panel A.	: Males			
Schooling	-0.016	-0.018	0.003	0.002	0.012	0.003	0.036	0.033
-	(0.014)	(0.014)	(0.012)	(0.012)	(0.082)	(0.080)	(0.031)	(0.030)
Ν	350,865	350,865	350,865	350,865	108,832	108,832	228,306	228,306
				Panel B:	Females			
Schooling	-0.007	-0.013	0.006	0.004	-0.103	-0.079	-0.027	-0.006
-	(0.016)	(0.013)	(0.016)	(0.012)	(0.084)	(0.064)	(0.043)	(0.036)
Ν	362,879	362,879	362,879	362,879	174,115	174,115	258,930	258,930
Birth FE	YES	YES	YES	YES	YES	YES	YES	YES
Municipal. FE	YES	YES	YES	YES	YES	YES	YES	YES
Spousal schooling	NO	YES	NO	YES	NO	YES	NO	YES

Table 6. Results for married males and females. Outcome is spousal stock market participation, risky market participation, (log) share of stocks and risky assets.

Notes: The table shows the effect of schooling on spousal stock market participation, risky market participation, and the (log)share of stocks and risky holdings out of total financial wealth for the sample of married men and women. Estimates are shown with and without controls for spousal education, where both own and spousal education are instrumented with reform exposure. Robust standard errors in parentheses. Standard errors clustered at the municipality level. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

	(1	)	(2		(3)		(4	-)	
	Stock market	participation	Risky market participation		(log)share	(log)share of stocks		(log)share of risky assets	
	OLS	IV	OLS	IV	OLS	IV	OLS	IV	
				Panel A	: Fathers				
Schooling	0.018	0.002	0.023	-0.005	0.011	-0.039	0.010	-0.013	
-	(0.000)***	(0.009)	(0.000)***	(0.010)	(0.001)***	(0.084)	(0.001)***	(0.039)	
Ν	713,886	713,886	713,886	713,886	147,730	147,730	351,909	351,909	
				Panel B	: Mothers				
Schooling	0.018	-0.013	0.024	-0.022	0.006	-0.000	0.007	0.025	
-	(0.000)***	(0.014)	(0.000)***	(0.015)	(0.002)***	(0.105)	(0.001)***	(0.051)	
Ν	920,148	920,148	920,148	920,148	199,629	199,629	458,204	458,204	
Birth FE	YES	YES	YES	YES	YES	YES	YES	YES	
Municipal. FE	YES	YES	YES	YES	YES	YES	YES	YES	

# Table 7. Results for children's outcomes

Notes: The table shows the effect of parental schooling on the children's stock market participation, risky market participation, and the (log)share of stocks and risky holdings out of total financial wealth. All models include birth cohort fixed effects and municipality fixed effects. Robust standard errors in parentheses. Standard errors clustered at the municipality level. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

	Stock market	Risky market	(log)share of	(log)share	(log)
	participation	participation	stocks	of risky assets	Financial wealth
		Par	nel A: Males		
Schooling	0.012	0.004	0.087	0.049	0.115
C	(0.007)	(0.006)	(0.045)*	(0.026)*	(0.096)
Ν	662,096	662,096	275,389	417,957	662,096
		Pane	el B: Females		
Schooling	-0.012	-0.010	-0.046	0.006	-0.154
-	(0.012)	(0.009)	(0.103)	(0.037)	(0.181)
Ν	642,119	642,119	194,618	403,705	642,119
Birth FE	YES	YES	YES	YES	YES
Municipal. FE	YES	YES	YES	YES	YES
Financial wealth	YES	YES	YES	YES	

Table 8. Effect of education on financial decisions (controlling for financial wealth). Males and females.

Notes: All columns control for birth cohort fixed effects, municipality fixed effects. Columns (1)-(4) show instrumental variables estimates of the effect of schooling, controlling for log (financial wealth). Column (5) shows the instrumental variable estimates of the effect of schooling on log (financial wealth). Standard errors are clustered at the municipality level. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

	(1)	(2)	(3)	(4)
		Panel A	: Males	, <i>t</i>
Father's years of	0.013	-0.000	-0.000	0.000
schooling	(0.002)***	(0.000)	(0.000)	(0.000)
Mother's years of	0.005	-0.000	-0.000	-0.000
schooling	(0.001)***	(0.000)	(0.000)	(0.000)
N	662,096	662,096	662,096	662,096
		Panel B:	Females	
Father's years of	0.013	0.000	0.000	0.000
schooling	(0.002)***	(0.000)	(0.000)	(0.000)
Mother's years of	0.005	-0.000	-0.000	-0.000
schooling	(0.001)***	(0.000)	(0.000)	(0.000)
N	642,119	642,119	642,119	642,119
Birth FE	YES	YES	YES	YES
Municipal. FE	NO	YES	YES	YES
Region by cohort FE	NO	NO	YES	NO
Mun. trends	NO	NO	NO	YES

Table A1. Exogeneity of the reform. Predicting reform exposure by parental schooling.

Notes: The table shows estimates of the relationship between parental schooling and reform exposure. Column (1) shows results including only birth cohort fixed effects. Column (2) shows estimates from a specification including birth cohort and municipality fixed effects. In addition, columns (3)-(4) include: (3) region by cohort fixed effects, and (4) municipality-specific linear trends. All estimations include indicator variables for missing schooling for fathers and mothers. Missing schooling is replaced by municipality by cohort means of schooling. Standard errors are clustered at the municipality level. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1.

		Males	5				Females	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Stock market	Risky market	(log) share	(log) share	Stock	Risky	(log) share of	(log) share of
	participation	participation	of stocks	of risky	market	market	stocks	risky assets
				assets	participation	participation		
		Panel	A: Shrinking	window to coh	orts -5 to $+5$ y	ears around r	eform	
Schooling	0.019	0.023	0.086	0.031	-0.014	-0.016	0.077	0.054
	(0.009)**	(0.008)***	(0.054)	(0.028)	(0.015)	(0.016)	(0.093)	(0.041)
Birth FE	YES	YES	YES	YES	YES	YES	YES	YES
Municipal. FE	YES	YES	YES	YES	YES	YES	YES	YES
Observations	438,180	438,180	178,688	272,761	424,597	424,597	125,765	263,462
			Panel I	B: Dropping th	e three larges	t citites		
Schooling	0.021	0.016	0.077	0.027	-0.002	-0.002	0.039	0.022
	(0.007)***	(0.007)**	(0.040)*	(0.020)	(0.010)	(0.010)	(0.063)	(0.030)
Birth FE	YES	YES	YES	YES	YES	YES	YES	YES
Municipal. FE	YES	YES	YES	YES	YES	YES	YES	YES
Observations	579,156	579,156	242,135	369,479	560,947	560,947	168,694	354,925

Table A2. Robustness checks.

Notes: The table shows various robustness checks on the effect of schooling on the children's stock market participation, risky market participation, (log) shares of stocks and risky assets. Panel A only includes cohorts born 5 years prior to or 5 years after the cohorts that were first exposed to the reform. Panel B drops the three biggest cities. All models include birth cohort fixed effects and municipality fixed effects. Robust standard errors in parentheses. Standard errors clustered at the municipality level. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

	viales allu lei	iiuies.		
(1)	(2)	(3)	(4)	(5)
1999	2000	2001	2002	Average
				1999-2002
$P_{c}$	anel A: Stock	market part	ticipation, ma	les
0.008	0.018	0.018	0.022	0.017
(0.009)	(0.008)**	(0.008)**	(0.008)***	(0.008)**
664,606	662,096	659,572	656,710	655,912
Pa	nel A: Stock	market parti	cipation, fem	ales
-0.024	-0.018	-0.014	-0.010	-0.017
(0.019)	(0.017)	(0.015)	(0.015)	(0.016)
643,732	642,119	640,356	638,445	637,928
Pa	anel C: Risky	market par	ticipation, ma	les
0.016	0.012	0.011	0.012	0.012
(0.009)*	(0.008)	(0.008)	(0.008)	(0.008)
664,606	662,096	659,572	656,710	655,912
Pa	nel D: Risky	market parti	cipation, fem	ales
-0.029	-0.021	-0.017	-0.016	-0.022
(0.020)	(0.018)	(0.016)	(0.017)	(0.017)
643,732	642,119	640,356	638,445	637,928
YÉS	YES	YES	YES	YÉS
YES	YES	YES	YES	YES
	$\begin{array}{c} & & & Pa \\ & & & Pa \\ & & & 0.008 \\ & & & (0.009) \\ & & 664,606 \\ & & Pa \\ & & -0.024 \\ & & (0.019) \\ & & 643,732 \\ & & & Pa \\ & & & 0.016 \\ & & & (0.009)* \\ & & 664,606 \\ & & & Pa \\ & & -0.029 \\ & & & (0.020) \\ & & 643,732 \\ & & YES \end{array}$	1999 2000   Panel A: Stock 0.008 0.018   (0.009) (0.008)** 664,606 662,096   Panel A: Stock -0.024 -0.018 -0.017)   643,732 642,119 664,606 662,096   Panel C: Risky 0.016 0.012 (0.009)* (0.008)   664,606 662,096 Panel D: Risky -0.029 -0.021   (0.009)* (0.008) 664,606 662,096 Panel D: Risky   -0.029 -0.021 (0.020) (0.018) 643,732 642,119   YES YES YES YES YES	199920002001Panel A: Stock market part0.0080.0180.018(0.009)(0.008)**(0.008)**664,606662,096659,572Panel A: Stock market part-0.024-0.018-0.014(0.019)(0.017)(0.015)643,732642,119640,356Panel C: Risky market part0.0160.0120.011(0.009)*(0.008)(0.008)664,606662,096659,572Panel D: Risky market part-0.029-0.021-0.017(0.020)(0.018)(0.016)643,732642,119640,356YESYESYES	1999200020012002Panel A: Stock market participation, ma $0.008$ $0.018$ $0.018$ $0.022$ $(0.009)$ $(0.008)^{**}$ $(0.008)^{**}$ $(0.008)^{***}$ $664,606$ $662,096$ $659,572$ $656,710$ Panel A: Stock market participation, fema $-0.024$ $-0.018$ $-0.014$ $-0.010$ $(0.019)$ $(0.017)$ $(0.015)$ $(0.015)$ $643,732$ $642,119$ $640,356$ $638,445$ Panel C: Risky market participation, ma $0.016$ $0.012$ $0.011$ $0.012$ $(0.009)^{*}$ $(0.008)$ $(0.008)$ $(0.008)$ $664,606$ $662,096$ $659,572$ $656,710$ Panel D: Risky market participation, fema $-0.029$ $-0.021$ $-0.017$ $-0.016$ $(0.020)$ $(0.018)$ $(0.016)$ $(0.017)$ $643,732$ $642,119$ $640,356$ $638,445$ YESYESYESYESYES

Table A3. Effect of education on participation in stock markets and risky markets. Males and females.

Notes: Columns (1)-(5) show instrumental variables estimates of the effect of schooling from specifications including birth cohort and municipality fixed effects. Columns (1)-(4) show the results for data from 1999 to 2002, respectively. In Column (5), the outcome variable is the average of participation across the four years represented in Columns (1)-(4) and the sample is limited to those who are represented in the sample in all four years. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

	Cł	nildren of fath	ers	Children of mothers		
	Mean	Standard Deviation	Obs	Mean	Standard Deviation	Obs
Portfolio characteristics:						
Direct equity participation**	0.21	0.41	713886	0.22	0.41	920148
Direct equity share***	0.08	0.21	550067	0.09	0.22	712696
Risky market participation**	0.49	0.50	713886	0.50	0.50	920148
Risky share***	0.33	0.37	550067	0.33	0.37	712696
Demographic characteristics:						
Age in year 2006	32.05	4.21	713886	33.34	4.68	920148
Female	0.49	0.50	713886	0.49	0.50	920148

Table A4. Summary Statistics for children.

Notes: \* A dummy, taking a value of 1 if the individual participates. \*\* Conditional on participation. \*\*\* A dummy, taking a value of 1 if the child is a female.

	(1)	(2)	(3)	(4)	(5)
	Earnings	Stock market	Risky market	(log)share of	(log)share
		participation	participation	stocks	of risky
					assets
Schooling	0.030	0.015	0.008	0.114	0.049
	(0.012)***	(0.008)*	(0.008)	(0.053)**	(0.029)*
Ν	658,479	658,479	658,479	274,029	416,117
Schooling	-0.039	-0.015	-0.018	0.017	0.024
U	(0.030)	(0.015)	(0.016)	(0.094)	(0.036)
Ν	638,412	638,412	638,412	193,488	401,680
Birth FE	YES	YES	YES	YES	YES
Municipal. FE	YES	YES	YES	YES	YES
Earnings		YES	YES	YES	YES

Table A5. Effect of education on financial decisions (controlling for earnings). Males and females.

Notes: All columns control for birth cohort fixed effects, municipality fixed effects. Column (5) shows the instrumental variable estimates of the effect of schooling on earnings. Columns (2)-(5) show instrumental variables estimates of the effect of schooling, controlling for log(earnings). Standard errors are clustered at the municipality level. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

Wealth cutoff	Full sample	99th percentile	95th percentile	90th percentile	75th percentile	
		Panel A: Males				
Schooling	0.115	0.131	0.171	0.181	0.158	
	(0.096)	(0.092)	(0.084)**	(0.080)**	(0.083)*	
Ν	662,096	655,475	628,991	595,886	496,572	
		Panel A: Females				
Schooling	-0.154	-0.133	-0.088	0.027	0.030	
	(0.181)	(0.159)	(0.141)	(0.121)	(0.118)	
Ν	642,119	635,697	610,013	577,906	481,589	
Birth FE	VES	VES	YES	YES	YES	
BIRIN FE	YES	YES	YES	YES	I ES	
Municipal. FE	YES	YES	YES	YES	YES	

Table A6. Effect of education on log (financial wealth) for different wealth cutoffs. M	Aales and
females.	

Notes: All columns control for birth cohort fixed effects, municipality fixed effects. Columns (1)-(5) show instrumental variables estimates of the effect of schooling on financial wealth for different wealth cutoffs. Standard errors are clustered at the municipality level. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

UCD CENTRE FOR ECONOMIC RESEARCH – RECENT WORKING PAPERS

WP14/07 Alan Fernihough, Cormac Ó Gráda and Brendan M Walsh: 'Mixed Marriages in Ireland A Century Ago' March 2014

<u>WP14/08</u> Doireann Fitzgerald and Stefanie Haller: 'Exporters and Shocks: Dissecting the International Elasticity Puzzle' April 2014

WP14/09 David Candon: 'The Effects of Cancer in the English Labour Market' May 2014

<u>WP14/10</u> Cormac Ó Gráda and Morgan Kelly: 'Speed under Sail, 1750–1850' May 2014

<u>WP14/11</u> Johannes Becker and Ronald B Davies: 'A Negotiation-Based Model of Tax-Induced Transfer Pricing' July 2014

<u>WP14/12</u> Vincent Hogan, Patrick Massey and Shane Massey: 'Analysing Match Attendance in the European Rugby Cup' September 2014

<u>WP14/13</u> Vincent Hogan, Patrick Massey and Shane Massey: 'Competitive Balance: Results of Two Natural Experiments from Rugby Union' September 2014 <u>WP14/14</u> Cormac Ó Gráda: 'Did Science Cause the Industrial Revolution?' October 2014

<u>WP14/15</u> Michael Daly, Liam Delaney, Orla Doyle, Nick Fitzpatrick and Christine O'Farrelly: 'Can Early Intervention Policies Improve Well-being? Evidence from a randomized controlled trial' October 2014

<u>WP14/16</u> Thérèse McDonnell and Orla Doyle: 'Maternal Employment, Childcare and Childhood Overweight during Infancy' October 2014

<u>WP14/17</u> Sarah Parlane and Ying-Yi Tsai: 'Optimal Sourcing Orders under Supply Disruptions and the Strategic Use of Buffer Suppliers' October 2014

<u>WP14/18</u> Morgan Kelly and Cormac Ó Gráda: 'Ready for Revolution? The English Economy before 1800' November 2014

<u>WP14/19</u> Johannes Becker, Ronald B Davies and Gitte Jakobs: 'The Economics of Advance Pricing Agreements' November 2014

<u>WP14/20</u> David Madden: 'Bridging the Gaps: Inequalities in Childrens' Educational Outcomes in Ireland ' November 2014

<u>WP14/21</u> Ronald B Davies, Julien Martin, Mathieu Parenti and Farid Toubal: 'Knocking on Tax Haven's Door: Multinational Firms and Transfer Pricing' December 2014

<u>WP15/01</u> Ronald B Davies, Benjamin H Liebman and Kasaundra Tomlin: 'I've Been Everywhere (Except Mexico): Investor Responses to NAFTA's Cross-Border Trucking Provisions' January 2015

<u>WP15/02</u> David Madden and Michael Savage: 'Which Households Matter Most?' Capturing Equity Considerations in Tax Reform via Generalised Social Marginal Welfare Weights' February 2015

<u>WP15/03</u> Ronald B Davies, Rodolphe Desbordes and Anna Ray: 'Greenfield versus Merger & Acquisition FDI: Same Wine, Different Bottles?' February 2015

<u>WP15/04</u> David Candon: 'Are Cancer Survivors who are Eligible for Social Security More Likely to Retire than Healthy Workers? Evidence from Difference-in-Differences' February 2015

<u>WP15/05</u> Morgan Kelly and Cormac Ó Gráda: 'Adam Smith, Watch Prices, and the Industrial Revolution' March 2015

<u>WP15/06</u> Kevin Denny: Has Subjective General Health Declined with the Economic Crisis? A Comparison across European Countries' March 2015 <u>WP15/07</u> David Candon: 'The Effect of Cancer on the Employment of Older Males: Attenuating Selection Bias using a High Risk Sample' March 2015 <u>WP15/08</u> Kieran McQuinn and Karl Whelan: 'Europe's Long-Term Growth Prospects: With and Without Structural Reforms' March 2015

UCD Centre for Economic Research

Email <u>economics@ucd.ie</u>